Author's response to reviews

Title: Iranian Cancer Patients' Perception of Spirituality: A Qualitative Content Analysis study

Authors:

Mojgan Rahnama (rahnamamo@gmail.com)
Masoud Fallahi Khoshknab (msflir@yahoo.com)
Sadat Sayed Bagher Maddah (maddah.ssb@gmail.com)
Fazlollah Ahmadi (ahmadif@modares.ac.ir)

Version: 5 Date: 13 September 2012

Author's response to reviews: see over
To dear editorial board of the journal
Thanks for your consideration,

**Competing interest**
The authors declare that they have no competing interest.

**Authors’ contributions**
MR, MFK and SSBM designed the study; MR collected data; MR, MFK, SSBM and FA contributed to data analysis, interpretation of findings and drafting of the manuscript. Both authors read and approved the final manuscript.

Mozhgan RAHNAME (MR)
Masoud FALLAH KHOSHKNAB (MFK)
Sadat Seid Bagher MADDAH (SSBM)
Fazlollah AHMADI (FA)

**Authors’ information (Authors’ Details)**
1) Department of Nursing, University of Social Welfare and Rehabilitation Sciences, Tehran, Iran. 2) Department of Nursing, Tarbiat Modares University, Tehran, Iran.

Rahnama